

GOPEN ACCESS

Citation: Musa M.A, Nakamura R, Hena A, Varikuti S, Nakhasi HL, Goto Y, et al. (2019) Lymphocytes influence *Leishmania major* pathogenesis in a strain-dependent manner. PLoS Negl Trop Dis 13 (11): e0007865. https://doi.org/10.1371/journal.pntd.0007865

Editor: Syamal Roy, CSIR-Indian Institute of Chemical Biology, INDIA

Received: June 18, 2019

Accepted: October 22, 2019

Published: November 18, 2019

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0007865

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

RESEARCH ARTICLE

Lymphocytes influence *Leishmania major* pathogenesis in a strain-dependent manner

Md. Abu Musa^{1,2,3}, Risa Nakamura^{1,2,3}, Asma Hena^{2,3}, Sanjay Varikuti⁴, Hira L. Nakhasi⁵, Yasuyuki Goto⁶, Abhay R. Satoskar⁴, Shinjiro Hamano^{1,2,3}*

1 Doctoral Leadership Program, Graduate School of Biomedical Sciences, Nagasaki University, Nagasaki, Japan, 2 Department of Parasitology, Institute of Tropical Medicine (NEKKEN), Nagasaki University, Nagasaki, Japan, 3 The Joint Usage/Research Center on Tropical Disease, Institute of Tropical Medicine (NEKKEN), Nagasaki University, Nagasaki, Japan, 4 Department of Pathology, Ohio State University, Columbus, Ohio, United States of America, 5 Division of Emerging and Transfusion Transmitted Diseases, CBER, FDA, Silver Spring, Maryland, United States of America, 6 Laboratory of Molecular Immunology, Graduate School of Agricultural and Life Sciences, The University of Tokyo, Bunkyo-ku, Tokyo, Japan

* shinjiro@nagasaki-u.ac.jp

Abstract

Cutaneous leishmaniasis (CL) is the most common form of leishmaniasis and is caused by several species of Leishmania parasite. Clinical presentation of CL varies from a self-healing infection to a chronic form of the disease determined by the virulence of infecting Leishmania species and host immune responses to the parasite. Mouse models of CL show contradictory roles of lymphocytes in pathogenesis, while acquired immune responses are responsible for host protection from diseases. To reconcile the inconclusive roles of acquired immune responses in pathogenesis, we infected mice from various genetic backgrounds with two pathogenic strains of Leishmania major, Friedlin or 5ASKH, and assessed the outcome of the infections. Our findings showed that the genetic backgrounds of L. major determine the impact of lymphocytes for pathogenesis. In the absence of lymphocytes, L. major Friedlin induced the lowest inflammatory reaction and pathology at the site of infection, while 5ASKH infection induced a strong inflammatory reaction and severe pathology. Lymphocytes ameliorated 5ASKH mediated pathology, while it exacerbated pathology during Friedlin infection. Excess inflammatory reactions, like the recruitment of macrophages, neutrophils, eosinophils and production of pro-inflammatory cytokines, together with uncontrolled parasite growth in the absence of lymphocytes during 5ASKH infection may induce severe pathology development. Taken together our study provides insight into the impact of differences in the genetic background of Leishmania on CL pathogenesis.

Author summary

Cutaneous leishmaniasis is caused by different species and sub-species of the intracellular parasite *Leishmania*. It is prevalent mainly in tropical and sub-tropical parts of the world. Disease manifestations range from self-healing cutaneous lesions to chronic form of the disease, depending on the infecting species of *Leishmania* and host immune protection. The mechanisms of pathogenesis are largely unknown. Lymphocytes play a central role in



Funding: This work was supported by the Joint Usage/Research Center on Tropical Disease, Institute of Tropical Medicine (NEKKEN), Nagasaki University (to RN, SH), the Global Health Innovative Technology (GHIT) Fund, Japan [G2015-115] (to AS, HN, SH), the Global Leadership Program, Nagasaki University (to AM, RN, SH) and JSPS KAKENHI [18H02649](to YG). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

the protection against *Leishmania* infection; however, their role in pathogenesis is poorly defined. Experimental infection studies showed the inconsistent role of lymphocytes in pathogenesis. Here, we compared disease outcomes in mice infected with different strains of *Leishmania major*, either Friedlin or 5ASKH. The pathogenesis caused by *L. major* 5ASKH infection was suppressed by the lymphocytes, while it was augmented by the lymphocytes during *L. major* Friedlin infection. Thus we found that the influence of lymphocytes in pathogenesis was determined by the genetic background of the parasites.

Introduction

Leishmaniasis is a group of diseases caused by protozoan parasites belonging to *Leishmania* species. Leishmaniasis threatens more than 350 million people from 98 countries in tropical and subtropical regions of the world with an estimated 0.7–1 million new cases and 20,000 to 30,000 deaths annually [1]. The disease can be categorized into three major forms: cutaneous, mucocutaneous and visceral leishmaniasis [2]. Cutaneous leishmaniasis (CL) is the most common form of the disease. Clinical manifestations of CL vary from asymptomatic to a non-healing chronic form of the disease determined by the infecting *Leishmania* species and by the host immune response to the parasite [3, 4]. The biological processes that may help to explain why some cases are asymptomatic and others show a variety of clinical pathologies, across the spectrum of self-healing to severe forms of leishmaniasis, are ill-defined. The diverse clinical manifestations of CL demand a thorough investigation of the pathophysiology of the infection in varying host defenses and parasite virulence.

Mouse models of leishmaniasis have been extensively studied to understand and characterize the host-parasite interactions during infections [5]. In mouse models of CL, the mechanism of resistance to *Leishmania major* infection has been well established using genetically resistant C57BL/6 mice that reproduce the self-cure disease outcomes typically seen in humans [3]. Resistance to *L. major* infection is mainly mediated by the CD4⁺ and CD8⁺ T lymphocytes, represented by the induction of Th1 immune response and IFN- γ dependent killing of the parasite by macrophages [5, 6]. Conversely, *L. major* infection drives BALB/c mice towards a Th2 response that allows for parasite replication and persistence [5].

While the role of the acquired immune response in protection against leishmaniasis has been well documented, its role in pathogenesis is not fully understood. Experimental CL studies showed the discrepant role of T/B lymphocytes in disease pathology. Some studies showed that lesion development was independent of T/B cells in CL models using *L. major* [7] or *Leishmania amazonensis* [8]. Other studies showed that lesion development caused by *L. major* [9, 10] or *L. amazonensis* [11] infection was delayed and less severe in the absence of T/B cells. Soong *et al.* showed that CD4⁺ T lymphocytes are indispensable for pathogenesis during *L. amazonensis* infection [12]. In these studies, different species or isolates of *Leishmania* parasite were used. The discrepant roles of T/B cells in pathogenesis may be due to the differences in parasite species or strains. It is reasonable to hypothesize that the genetic background of the *Leishmania* strain is crucial to determine the variable influence of lymphocytes for pathology development.

To investigate our hypothesis, we infected mice of various background with two strains of *L. major* with different degrees of pathogenicity, namely strains Friedlin and 5ASKH, and measured the disease outcome. Both *L. major* Friedlin and 5ASKH infections were controlled in resistant C57BL/6 mice with transient lesion development and limiting parasite burden. Susceptible BALB/c mice showed progressive lesion development with uncontrolled parasite

growth. In the absence of T/B lymphocytes, 5ASKH strain caused more progressive lesion development, while Friedlin strain caused limited lesion development whilst maintaining similar levels of parasite burden to 5ASKH strain. Our results indicated that lymphocytes suppressed pathogenesis during 5ASKH infection, while the presence of lymphocytes during Friedlin infection upregulated pathogenesis. We conclude that the genetic background of *L. major* determines the distinct influence of lymphocytes for the pathogenesis of this organism.

Materials and methods

Ethics statement

The study protocol was approved by the Committee for Ethics on Animal Experiments (approval number 1505181226, 1505181227) and Recombinant DNA experiments (1403041262) in Nagasaki University. All of the studies were conducted under the guidelines for animal experiments, Nagasaki University and according to Japanese law for Humane Treatment and Management of Animals (Law No. 105 dated 19 October 1973 modified on 2 June 2006).

Animals

Six to eight weeks old female mice were used in all experiments. In this study, we included wild-type (WT) and nude BALB/c, WT and recombination activating gene 2 (Rag2) knockout (KO) C57BL/6, and NOD-SCID mice. WT BALB/c (BALB/cCrSlc) and C57BL/6 (C57BL/ 6JJmsSlc) mice were purchased from SLC, Japan. Nude BALB/c (CAnN.Cg-Foxn1^{nu}/CrlCrlj) and NOD-SCID (NOD.CB17-Prkdc scid/J) mice were purchased from Charles River Laboratories, Japan. Rag2 KO C57BL/6 (B6(Cg)-Rag2tm1.1Cgn/J) mice were purchased from the Jackson Laboratory. Experimental mice were maintained under specific pathogen-free conditions at the Animal Facilities of Nagasaki University. Experimental animals were fed with standard rodent pellets supplemented with grain and given water ad libitum. Throughout this study, the animals were housed in a climate-controlled (23±2°C; relative humidity, 60%) and photoperiod controlled (12-hours light-dark cycles) animal quarters. Only healthy mice were included in the study.

Leishmania parasite and infection

L. major clone V9 (MHOM/IL/80/Friedlin) and L. major (MHOM/SU/73/5-ASKH) strain abbreviated as L. major Friedlin and L. major 5ASKH respectively were used in this study. The parasite culture was maintained in medium 199 (Gibco) supplemented with 10% heat-inactivated fetal bovine serum (CCB, Japan), 100 U/ml penicillin and 100 µg/ml streptomycin (Gibco) at 26°C in a biological incubator. Cultures were passaged to fresh medium at a 50-fold dilution in every 2-3 days. Stationary phase promastigotes were harvested and suspended in $1 \times PBS$. Mice were infected with 5×10^6 stationary phase promastigotes of L. major Friedlin or 5ASKH by subcutaneous injections into their right hind footpad. Following infection, footpad thickness was measured weekly using an analog caliper. Footpad swelling was calculated by subtracting the thickness of the uninfected counter footpad from the thickness of the infected footpad. Parasite burdens in the footpads and popliteal lymph nodes (pLN) were quantified at 6 weeks post-infection. Parasite burdens in the footpads and draining lymph nodes were quantified by either limiting dilution assay (LDA) or homogenizing tissue in 10 ml of complete medium 199 and incubated in 25cm² culture flasks at 26°C for the detection of parasites transformed in vitro from amastigotes to motile promastigotes. At 3 days of culture, promastigotes were counted microscopically. For LDA tissues were homogenized in 1 ml complete medium

199 following 2 fold serial dilution across 96-well culture plates. Culture plates were incubated at 26°C. At 1-week culture plates were observed under a microscope for the presence of motile promastigotes.

Isolation of cells from the footpad of mouse

Mice were euthanized and the footpads have collected aseptically. The footpads were cut tangentially to the bone and mechanically disrupted with surgical scissors in 5 ml cold RPMI medium supplemented with 1 mg/ml collagenase (WAKO, Japan) and 1 mg/ml DNase (Roche Diagnostics, Japan). Disrupted footpads were then incubated for 30 min in a 37°C water bath with continuous shaking. Following incubation, the reactions were stopped by adding cold RPMI medium with 2.5 mM EDTA and strained through a 70 μ m cell strainer. The cell suspensions were centrifuged at 440g for 6 min at 4°C and re-suspended in complete RPMI medium supplemented with 10% heat-inactivated fetal bovine serum, 100 U/ml penicillin and 100 μ g/ml streptomycin.

Flow cytometric analysis

Footpad cells were stained separately with 2 sets of fluorescence-conjugated antibody cocktail. The first set of antibody cocktail consisted of PE-conjugated anti-CD86, APC-conjugated anti-MHC class-II, FITC-conjugated anti-CD11b, PE-Cy7-conjugated anti-F4/80, APC-Cy7-conjugated anti-CD45, and PerCp-Cy5.5-conjugated anti-CD11c. The second set of the antibody included PE-conjugated anti-SiglecF, APC-conjugated anti-Gr1, FITC-conjugated anti-CD11b, PE-Cy7-conjugated anti-Gr1, FITC-conjugated anti-CD11b, PE-Cy7-conjugated anti-F4/80, APC-Cy7-conjugated anti-CD15, and PerCp-Cy5.5-conjugated anti-F4/80, APC-Cy7-conjugated anti-CD45, and PerCp-Cy5.5-conjugated anti-CD45, and PerCp-Cy5.5-conjugated anti-SiglecF, APC-cy7-conjugated anti-CD45, and PerCp-Cy5.5-conjugated anti-CD45, and PerCp-Cy5.5-conjugated anti-CD45, and PerCp-Cy5.5-conjugated anti-SiglecF, APC-cy7-conjugated anti-CD45, and PerCp-Cy5.5-conjugated anti-NKp46 antibody. Cells were then analyzed on a FACSVerse[™] (BD Bio-science, NJ, USA). Data were analyzed with FlowJo software v10.

Ex-vivo cytokine analysis

Cells isolated from mouse footpad were plated at the concentration of 5×10^5 cells per well in 96-well culture plates in 200 µl RPMI medium (supplemented with 10% heat-inactivated fetal bovine serum, 100 U/ml penicillin and 100 µg/ml streptomycin). After 72h of PMA (40 ng/ml)/Ionomycin (4 µg/ml) stimulation, culture supernatants were stored at -30°C. Concentrations of IFN- γ , TNF- α , IL-12 p40, and IL-10 were measured by sandwich ELISA as per manufacturer (R&D Systems) instructions.

Quantitative RT-PCR

Mice footpads were homogenized with 1 ml TRIzol (Thermo Fisher Scientific) and φ 1.0 stainless steel beads in the 2 ml tube using Micro Smash MS100R (TOMY, Tokyo, Japan) at 4°C using tissue homogenizer (Tomy, Micro Smash MS-100). RNA was isolated using Trizol Reagent according to manufacturer instructions. RNA yield was measured by a spectrophotometer (Beckman Coulter DU-730). We found 30–53 µg RNA from each sample, and 2 µg of total RNA was used as the template for the synthesis of 20 µl cDNA. Synthesized cDNA was analyzed for mouse TNF- α , IL-1 β , IL-6, IL-12 p40, inducible nitric oxide synthase (iNOS), and arginase 1 by reverse transcriptase real-time PCR. PCR method and primer sequences were described before [13]. Briefly real-time polymerase chain reaction (PCR) assay was carried out using 1 µl of cDNA as the template, 10 µl of SYBR Select Master Mix (Thermo Fisher Scientific) and primers on the ABI Prism 7000 Sequence Detection System (Thermo Fisher Scientific). Data were analyzed by 2– $\Delta\Delta$ Ct methods and normalized by GAPDH. The thermal cycling conditions for the PCR were 94°C for 10 min, followed by 40 cycles of 94°C for 15 s and 60°C for 1 min.

Statistical analysis

All statistical analysis was performed using GraphPad Prism Software Version-5. To check data distribution we had performed the Shapiro-Wilk normality test. Statistical analysis of the lesion kinetics was performed with two-way ANOVA to test differences between groups. In all cases, a P-value of less than 0.05 was considered statistically significant.

Results

Parasite strain-dependent influence of T lymphocytes on pathogenesis in susceptible BALB/c mice

To characterize the role of T lymphocytes and *L. major* strains on pathogenesis, we infected susceptible BALB/c WT and nude mice with 5×10^6 stationary phase promastigotes of either *L. major* strain, Friedlin or 5ASKH. Following infections, the lesions were measured weekly, until 6 weeks post-infection. The parasite burden in footpads and pLN were also measured at 6 weeks post-infection, at the end of the experiment. Both *L. major* Friedlin and 5ASKH infection showed progressive lesion development in WT mice (**Fig 1A**), where three of eight mice infected with 5ASKH developed ulcerative lesions compared with one of nine for Friedlin-infected mice (**Table 1**). In nude mice, *L. major* 5ASKH infection caused more severe ulcerative lesions in all 10 mice compared to 3/8 in WT mice (**Fig 1B and Table 1**). Conversely, *L. major* Friedlin infection caused minimal lesion development with no ulcer formation in all 11 nude mice and 1/9 in WT mice (**Fig 1B and Table 1**). It is worth noting that both Friedlin and 5ASKH showed a similarly high level of parasite burden in the footpads and pLN not only in nude mice but also in WT mice (**Figs 1C and S1**). These results showed that T lymphocytes reduced immunopathology during 5ASKH infection, while T lymphocytes exacerbated lesion development during Friedlin infection.

Parasite strain-dependent influence of T/B lymphocytes on pathogenesis in resistant C57BL/6 mice

Next, we examined the *L. major* strain-dependent influence of lymphocytes for lesion development in resistant C57BL/6 mice that allows transient *L. major* infection. WT and Rag2 KO C57BL/6 mice were infected with either *L. major* Friedlin or 5ASKH. The WT mice showed transient non-ulcerative lesion development during infection with both *L. major* strains (Fig 2A and 2B) and controlled parasite growth (Fig 2C). In Rag2 KO mice, which lack both conventional T/B lymphocytes, 5ASKH infection caused delayed but progressive and non-ulcerative lesion development (Fig 2A and 2B) despite a similar level of parasite burden in Friedlin- or 5ASKH-infected footpads and pLN (Figs 2C and S2).

L. major Friedlin and 5ASKH caused minimum or severe pathology respectively in NOD-SCID mice

Lack of pathogenesis by Friedlin strain in severe immunodeficient condition was also confirmed in NOD-SCID mice. NOD-SCID mice lack T/B cells and also defects in functions of NK, NKT, macrophage, and dendritic cells. In NOD-SCID mice, there was no difference in parasite burden between Friedlin and 5ASKH infected footpad and pLN (S3C Fig). However,



Fig 1. Parasite strain-dependent influence of T lymphocytes on pathogenesis in susceptible BALB/c mice. BALB/c WT and nude mice were infected with 5×10⁶ stationary phase promastigotes of *L. major* Friedlin or 5ASKH subcutaneously into the hind footpad. (**A**) Footpad swelling after *L. major* infection. Each group contains 8–11 mice. All the data are from normal distribution except data of BALB/c mice at 1 and 3 weeks and BALB/c-nude mice at 2 weeks post-infection of *L. major* Friedlin have deviated from normal distribution. Data deviated from normal distribution were excluded from statistical comparison (**B**) Representative photographs of *L. major*-infected footpads at 6 weeks post-infection. (**C**) Parasite burden in footpads at 6 weeks post-infection. Each mouse footpad was homogenized in 10 ml culture media and cultured for 3 days then viable promastigotes were counted microscopically. Each group contains four-six mice. Symbols: BALB/c mice infected with *L. major* Friedlin black circle or 5ASKH red triangle; Data are the mean ± SEM. Statistical analyses for lesion size by ANOVA. *†¶# means P<0.05 between BALB/c-Friedlin×Nude-5ASKH: *; Nude-Friedlin×Nude-5ASKH: *; MALB/c-Friedlin×Nude-5ASKH: *; MaLB/c-Friedlin×Nude-Friedlin×Nude-5ASKH: *.

https://doi.org/10.1371/journal.pntd.0007865.g001

L. major 5ASKH showed non-ulcerative and progressive lesion development, but Friedlin showed almost no lesion development (Table 1 and S3A and S3B Fig).

Mice	Ulcer formation % (number)	
	Friedlin	5ASKH
BALB/c	11.1 (1/9)	37.5 (3/8)
BALB/c-nude	0 (0/11)	100 (10/10)
C57BL/6	0 (0/8)	0 (0/8)
C57BL/6-Rag2 KO	0 (0/5)	0 (0/3)
NOD-SCID	0 (0/8)	0 (0/8)

Table 1. Ulcer formation at 6 weeks post-infection with *L. major* Friedlin or 5ASKH.

Defective infiltration of immune cells to the site of *L. major* Friedlin infection

Our study showed that both parasite and host factors affected the clinical presentation of CL. Inflammatory responses are mediated by immune cells that control not only parasite growth but also contribute to the pathogenesis of CL [3, 14]. To characterize the inflammation induced by either *L. major* Friedlin or 5ASKH infections, we isolated immune cells from Rag2 KO-infected footpads at 4 weeks post-infection and analyzed these cells by flow cytometry. It is worth noting that, *L. major* Friedlin infection failed to recruit immune cells to the site of infection, while a 25 times higher number of cells were recruited to the footpads infected with *L. major* 5ASKH. Among the cells recruited to the site of *L. major* 5ASKH infection, macrophages (CD11b⁺F4/80⁺), neutrophils (CD11b⁺Gr1⁺), and eosinophils (CD11b⁺SiglecF⁺) were the major cell populations. In addition, *L. major* 5ASKH infected footpad also contained a higher proportion and number of activated macrophages (CD11b⁺F4/80⁺) compared to that with *L. major* Friedlin (**Figs** 3A and 3B and 4A and 4B and S4).

Lower activation of host immune responses by L. major Friedlin infection

Immune cells were isolated from the footpads of Rag2 KO mice 4 weeks after infection with *L. major* Friedlin or 5ASKH and were cultured for 72 h with or without PMA-ionomycin stimulation. Culture supernatants were analyzed for IFN- γ , TNF- α , IL-12, and IL-10 cytokines by ELISA. *L major* infection augmented the production of cytokines and *L. major* 5ASKH induced significantly higher cytokine production than Friedlin (Fig 5). After stimulation with PMA-ionomycin, a similar trend was observed (S5 Fig).

Discussion

The distinct impact of lymphocytes for the pathogenesis of CL was confirmed to be attributed to the genetic background of *L. major*. Our results clarified the contradictory findings of previous independent studies, in which lesion development by CL was either similar [7, 8], delayed or less pronounced [9, 10, 12] in the absence of T/B lymphocyte-mediated immune responses compared to wild-type mice. The *L. major* 5ASKH infection model was consistent with reports that lesion development by either *L. major* (MHOM/UZ/91/PM2) or *L. amazonensis* (MPRO/ BR/72/M1845) was similar or less in the presence of lymphocytes [7, 8]. The *L. major* Friedlin infection model also explains the previous report that delayed or no lesion development in the absence of T/B lymphocytes by *L. major* (WHOM/IR/-/173) [9, 10] or *L. amazonensis* (MHOM/BR/77/LTB0016) [12] infection. Thus, the impact of lymphocytes for pathogenesis can be determined by the genetic background of *L. major*. In this study, the genetic background of *L. major* or or less that study. We compared not by the strains of *L. major* we had no comparison with other strains of *L. major* or



Fig 2. Parasite strain-dependent influence of T/B lymphocytes on pathogenesis in resistant C57BL/6 mice. C57BL/ 6 WT and Rag2 KO mice were infected with 5×10⁶ stationary phase promastigotes of *L. major* Friedlin or 5ASKH subcutaneously into the hind footpad. (A) Footpad swelling after *L. major* infection. All the data are from normal distribution except data of C57BL/6 mice at 1 week post-infection of *L. major* Friedlin have deviated from normal distribution. Data deviated from normal distribution were excluded from statistical comparison. Symbols: WT mice infected with *L. major* Friedlin blue circle or 5ASKH green triangle, Rag2 KO mice infected with *L. major* Friedlin black circle or 5ASKH red triangle; 3–10 mice per group. (B) Representative photographs of a *L. major*-infected footpad at 6 weeks post-infection. (C) Parasite burden at 6 weeks post-infection. Each mouse footpad was homogenized in 10 ml culture media and cultured for 3 days then viable promastigotes were counted microscopically. Three-five mice per group. Data are the mean ± SEM. Results are representative of two independent experiments with similar outcomes. *†¶ means P<0.05 between WT-Friedlin×WT-5ASKH: *; Rag2KO-Friedlin×Rag2KO-5ASKH: †; WT-Friedlin×Rag2KO-Friedlin: ¶ and WT-5ASKH×Rag2KO-5ASKH: #.

different species of *Leishmania*. Further immunological analyses are needed to inquire into the differential lymphocyte response towards different *Leishmania* strains and species.



Fig 3. *L. major* Friedlin infection failed to augment the recruitment of macrophages in Rag2 KO mice. Rag2 KO C57BL/6 mice were infected with 5×10^6 stationary phase promastigotes of *L. major* Friedlin or 5ASKH subcutaneously into the hind footpad. At 4 weeks post-infection, cells from the mouse footpads were isolated and analyzed by flow cytometry. The total number (A) and percentages (B) of the cell populations are shown. Four-five mice per group. Data are the mean \pm SEM.

In Rag2–deficient C57BL/6 mice, 5ASKH infection caused progressive lesion development. In the early phase of its infection, the degree of footpad swelling was delayed in the absence of T/B lymphocytes indicating that even during 5ASKH infection, lymphocytes could be attributed to lesion formation to some extent. In fact, 5ASKH-infected footpads of WT mice showed a trend of higher expression of pro-inflammatory molecules, such as IL-6, TNF- α , IL-1 β , IL-12p40 and iNOS compared with Friedlin-infected footpads (S6 Fig). The preceding studies also independently showed delayed lesion development after *L. major* (WHO strain WHOM/IR/-/173) infection in C.B-17 SCID mice compared with BALB/c mice [9, 10].

Pathogenesis can be caused by either the direct tissue-damaging effect of the parasites, parasite-induced immune-mediated skin inflammation or their combined effects. Belkaid *et al.* showed that lesion development starts after the parasite burden reaches its peak and that it coincides with leukocyte trafficking in resistant C57BL/6 mice [15], which emphasized the importance of parasite-induced inflammatory reactions for pathogenesis. We showed that *L. major* 5ASKH infection induced higher inflammatory reactions both in the presence and absence of T/B lymphocytes. Massive recruitment of immune cells, mainly macrophages, neutrophils and eosinophils, was observed at the site of 5ASKH infection even in the absence of lymphocytes. It is consistent with the prior study which showed that lesion development was associated with acute infiltration of leukocytes, primarily macrophages, neutrophils and





Fig 4. Defective infiltration of neutrophils, eosinophils, and NK cells during *L. major* Friedlin infection in Rag2 KO mice. Rag2 KO C57BL/6 mice were infected with 5×10^6 stationary phase promastigotes of *L. major* Friedlin or 5ASKH subcutaneously into the hind footpad. At 4 weeks post-infection, cells from the mouse footpads were isolated and analyzed by flow cytometry. The total number (A) and percentages (B) of the cell populations are shown. Four-five mice per group. Data are the mean \pm SEM.

eosinophils [15]. Immune cells recruited to the site of 5ASKH infection were also highly augmented with a larger number of CD86⁺-activated macrophages and a higher level of cytokine production was observed compared with Friedlin infection. The inflammatory cytokines are able to activate the macrophages for intracellular killing of the parasite. In wild-type mice, Tlymphocytes are known to be the major source of IFN- γ which is central for the protective immunity. In the absence of T/B lymphocytes, mice failed to control parasites. It indicates that T/B lymphocytes are critical for the control of the parasites, mainly through the production of sufficient amount of critical cytokines, like IFN- γ and TNF- α . In the absence of T/B lymphocytes, macrophages, dendritic cells, NK cells, and group 1 innate lymphoid cells may be one of the sources of IFN- γ , which is insufficient for the protection (**Fig 5**). It may explain why the immunodeficient mice infected with both *L. major* strains showed no difference in parasite burden. However, augmented inflammatory responses after 5ASKH infection may have caused the severe pathology observed. Strong inflammatory reactions at the site of *L. major* 5ASKH infection may explain the severe pathology observed even in the absence of T/B



Fig 5. Lower activation of host immune responses by *L. major* Friedlin infection. Rag2 KO C57BL/6 mice were infected with 5×10^6 stationary phase promastigotes of *L. major* Friedlin or 5ASKH subcutaneously into the hind footpad. At 4 weeks post-infection, cells from mouse footpads were isolated and cultured for 3 days. Culture supernatants were analyzed for (A) IFN- γ , (B) TNF- α , (C) IL-12p40, and (D) IL-10 cytokines by ELISA. Three–four mice per group. Data are the mean ± SEM. *, P<0.05.

lymphocytes. *L. major* Friedlin infection failed to induce inflammatory reactions, like recruitment of immune cells and pro-inflammatory cytokines production at the site of infection in the absence of T/B lymphocytes, which may explain the minimum pathology observed. The outcome of Friedlin infection in the immunodeficient mice was consistent with the reported no pathology associated with less infiltration of monocytes at the site of *L. amazonensis* infection in mice lacking T cells [12]. Although we compared inflammatory responses only at 4 weeks after *L. major* Friedlin/5ASKH infection, it is also important to compare early inflammatory responses to address the mechanisms responsible for the different disease outcomes. Different *Leishmania* parasites may have diverse molecules or strategies determining the impact of lymphocytes for immuno-pathogenesis. Comparative analysis showed a significant number of virulence factors or molecules differentially expressed by *Leishmania* isolates [16– 18]. We also analyzed only a few cytokines or host molecules after *L. major* infection. Further studies are required to identify the molecules and pathways involved.

CL has a wide spectrum of clinical manifestations depending on the host immunity and the infecting *Leishmania* species. CL is caused by several species of *Leishmania* parasite. In the new world, *L. mexicana*, *L. amazonensis*, *L. panamensis*, *L. brazilienesis*, *L. guyanensis*, *L per-uviana* and *L. venezuelensis* are the etiological agents of CL, whereas in the old world *L. tropica*,

L. killicki, L. aethiopica, L. arabica, L. infantum and *L. major* are responsible for CL [19, 20]. Many studies have been reported on CL, but few studies focused on the diversity of the clinical presentations. We extensively studied infection outcomes caused by two different strains of *L. major* in animal models with different genetic backgrounds and varying immune competency. *L. major* Friedlin and 5ASKH were originally isolated from patients with CL in the old world, Israel and Turkmenistan respectively, then adapted to the conditions in the Laboratory and established as the strains [21]. In immune-competent mice, we observed different infection outcomes in different host genetic backgrounds. Friedlin infection caused almost no ulceration in any genetic background studied, whereas 5ASKH infection caused more ulcerative lesions in T lymphocyte-deficient BALB/c mice compared with WT mice. Terabe *et al.* showed that T/B lymphocytes are a prerequisite for ulcer formation during *L. amazonensis* infection in BALB/c mice [8]. Therefore, particular *Leishmania* species may have distinct immunopathologies of ulcer formation.

Conclusion

Different species or strains of *Leishmania* were associated with a wide spectrum of diseases in human [19, 20, 22, 23] and mouse models. Previous studies showing that *L. major* LV39 caused a non-healing infection [24–26] whereas other *L. major* strain (MHOM/IL/81/FEBNI) caused healing infections in IL-4 deficient BALB/c mice [25–26]. Furthermore, it now appears that strains of *L. major* can cause variant pathogenesis even in the absence of lymphocytes. Lymphocytes differentially regulate pathogenesis during *L. major* Friedlin/5ASKH infection. *Leishmania* has mechanisms to evade host immune protection, thereby aiding survival inside the host [27]. *L. major* Friedlin may have a superior immune evasion mechanism compared with strain 5ASKH that allows it to survive within the host by eliciting minimum immune-mediated pathogenesis. Understanding the basis for the diversity in pathogenesis is important in determining how to apply our knowledge for the development of new approaches for the treatment of leishmaniasis.

Supporting information

S1 Fig. Parasite strain-dependent influence of T lymphocytes on pathogenesis in susceptible BALB/c mice. BALB/c WT and nude mice were infected with 5×10^6 stationary phase promastigotes of *L. major* Friedlin or 5ASKH subcutaneously into the hind footpad. Parasite burden in popliteal lymph nodes (pLN) at 6 weeks post-infection. Each mouse footpad was homogenized in 10 ml culture media and cultured for 3 days then viable promastigotes were counted microscopically. Each group contains 8–11 mice. Symbols: BALB/c mice infected with *L. major* Friedlin \circ or 5ASKH \triangle , BALB/c-nude mice infected with *L. major* Friedlin \bullet or 5ASKH \triangle Data are mean \pm SEM. (TIF)

S2 Fig. Parasite strain-dependent influence of T/B lymphocytes on pathogenesis in resistant C57BL/6 mice. C57BL/6 WT and Rag2 KO mice were infected with 5×10^6 stationary phase promastigotes of *L. major* Friedlin or 5ASKH subcutaneously into the hind footpad. Parasite burden in popliteal lymph nodes (pLN) at 6 weeks post-infection. Each mouse footpad was homogenized in 10 ml culture media and cultured for 3 days then viable promastigotes were counted microscopically. 3–9 mice per group. Symbols: wild-type mice infected with *L. major* Friedlin \circ or 5ASKH \triangle , Rag2 KO mice infected with *L. major* Friedlin \oplus or 5ASKH \blacktriangle . Data are mean \pm SEM. Results are representative of 2 independent experiments with a similar outcome. (TIF)

S3 Fig. *L. major* **Friedlin or 5ASKH infection in NOD-SCID mice.** NOD-SCID mice were infected with 5×10^6 stationary phase promastigotes of *L. major* Friedlin \bigcirc or 5ASKH \blacktriangle subcutaneously into the hind footpad. (A) Footpad swelling after *L. major* infection. Symbols: NOD-SCID mice infected with *L. major* Friedlin or 5ASKH; 8 mice per group. (B) Representative photographs of *L. major* infected footpad at 6 weeks post-infection. (C) Footpad parasite burden at 6 weeks post-infection. Each mouse footpad was homogenized in 10 ml culture media and cultured for 3 days then viable promastigotes were counted microscopically. 4–8 mice per group. Data are mean ± SEM. Results are representative of two independent experiments with a similar outcome.



S4 Fig. Flow cytometric analysis of mouse footpad cells. Rag2 KO C57BL/6 mice were infected with 5×10^6 stationary phase promastigotes of *L. major* Friedlin or 5ASKH subcutaneously into the hind footpad. At 4 weeks post-infection, cells were isolated from mouse footpads and analyzed by flow cytometry. Gating strategies for (A) macrophages; (B) neutrophils, eosin-ophils, and NK cells.

(TIF)

S5 Fig. Less activation of host immune responses by *L. major* Friedlin infection. Rag2 KO C57BL/6 mice were infected with 5×10^6 stationary phase promastigotes of *L. major* Friedlin or 5ASKH subcutaneously into the hind footpad. At 4 weeks post-infection cells were isolated from mouse footpads and stimulated with PMA-ionomycin for 3 days. Culture supernatants were analyzed for cytokines by ELISA. 3–4 mice per group. Data are mean \pm SEM. *, P<0.05. (TIF)

S6 Fig. *L. major* Friedlin infection induced a lower level of the immune response. C57BL/6 mice were infected with 5×10^6 stationary phase promastigotes of *L. major* Friedlin or 5ASKH subcutaneously into the hind footpad. At 4 weeks post-infection total RNA was isolated from mouse footpads and analyzed for mRNA of targeted molecules by real-time RT-PCR. (A) Footpad parasite burden at 4 weeks post-infection measured by limiting dilution assay. (B) mRNA expression in the infected footpad at 4 weeks post-infection. 3–4 mice per group. Data are mean \pm SEM.

(TIF)

Acknowledgments

We would like to offer special thanks to all members of the Department of Parasitology, Institute of Tropical Medicine (NEKKEN), Nagasaki University. We also thank Kate Fox, DPhil, from Edanz Group (www.edanzediting.com/ac) for editing a draft of this manuscript.

Author Contributions

Conceptualization: Md. Abu Musa, Shinjiro Hamano.

Data curation: Md. Abu Musa, Asma Hena.

Formal analysis: Md. Abu Musa, Risa Nakamura.

Funding acquisition: Yasuyuki Goto, Abhay R. Satoskar, Shinjiro Hamano.

Investigation: Md. Abu Musa, Risa Nakamura, Yasuyuki Goto, Shinjiro Hamano.

Methodology: Md. Abu Musa, Risa Nakamura, Sanjay Varikuti, Yasuyuki Goto, Shinjiro Hamano.

Resources: Yasuyuki Goto, Shinjiro Hamano.

Supervision: Risa Nakamura, Hira L. Nakhasi, Yasuyuki Goto, Abhay R. Satoskar, Shinjiro Hamano.

Validation: Risa Nakamura, Shinjiro Hamano.

Visualization: Md. Abu Musa.

Writing - original draft: Md. Abu Musa.

Writing – review & editing: Md. Abu Musa, Hira L. Nakhasi, Yasuyuki Goto, Shinjiro Hamano.

References

- 1. WHO. Leishmaniasis. Available from: https://www.who.int/leishmaniasis/en/.
- Sunter J, Gull K. Shape, form, function and Leishmania pathogenicity: from textbook descriptions to biological understanding. Open Biol. 2017; 7(9). <u>https://doi.org/10.1098/rsob.170165</u> PMID: 28903998; PubMed Central PMCID: PMC5627057.
- Nylen S, Eidsmo L. Tissue damage and immunity in cutaneous leishmaniasis. Parasite Immunol. 2012; 34(12):551–61. https://doi.org/10.1111/pim.12007 PMID: 23009296.
- Kaye P, Scott P. Leishmaniasis: complexity at the host-pathogen interface. Nat Rev Microbiol. 2011; 9 (8):604–15. https://doi.org/10.1038/nrmicro2608 PMID: 21747391.
- Sacks D, Noben-Trauth N. The immunology of susceptibility and resistance to Leishmania major in mice. Nat Rev Immunol. 2002; 2(11):845–58. https://doi.org/10.1038/nri933 PMID: 12415308.
- Von Stebut E, Ehrchen JM, Belkaid Y, Kostka SL, Molle K, Knop J, et al. Interleukin 1alpha promotes Th1 differentiation and inhibits disease progression in Leishmania major-susceptible BALB/c mice. J Exp Med. 2003; 198(2):191–9. <u>https://doi.org/10.1084/jem.20030159</u> PMID: <u>12860932</u>; PubMed Central PMCID: PMC2194079.
- Goto Y, Sanjoba C, Arakaki N, Okamoto M, Saeki K, Onodera T, et al. Accumulation of macrophages expressing MRP8 and MRP14 in skin lesions during Leishmania major infection in BALB/c and RAG-2 knockout mice. Parasitol Int. 2007; 56(3):231–4. <u>https://doi.org/10.1016/j.parint.2007.02.007</u> PMID: 17409015.
- Terabe M, Kuramochi T, Hatabu T, Ito M, Ueyama Y, Katakura K, et al. Non-ulcerative cutaneous lesion in immunodeficient mice with Leishmania amazonensis infection. Parasitol Int. 1999; 48(1):47–53. https://doi.org/10.1016/s1383-5769(98)00040-3 PMID: 11269325.
- Guy RA, Belosevic M. Response of scid mice to establishment of Leishmania major infection. Clin Exp Immunol. 1995; 100(3):440–5. https://doi.org/10.1111/j.1365-2249.1995.tb03719.x PMID: 7774053; PubMed Central PMCID: PMC1534477.
- Varkila K, Chatelain R, Leal LM, Coffman RL. Reconstitution of C.B-17 scid mice with BALB/c T cells initiates a T helper type-1 response and renders them capable of healing Leishmania major infection. Eur J Immunol. 1993; 23(1):262–8. https://doi.org/10.1002/eji.1830230141 PMID: 8419179.
- Velasquez LG, Galuppo MK, E DER, Brandao WN, Peron JP, Uliana SR, et al. Distinct courses of infection with Leishmania (L.) amazonensis are observed in BALB/c, BALB/c nude and C57BL/6 mice. Parasitology. 2016; 143(6):692–703. https://doi.org/10.1017/S003118201600024X PMID: 26892342.
- Soong L, Chang CH, Sun J, Longley BJ, Jr, Ruddle NH, Flavell RA, et al. Role of CD4+ T cells in pathogenesis associated with Leishmania amazonensis infection. J Immunol. 1997; 158(11):5374–83. PMID: 9164958.
- Mizobuchi H, Fujii W, Isokawa S, Ishizuka K, Wang Y, Watanabe S, et al. Exacerbation of hepatic injury during rodent malaria by myeloid-related protein 14. PLoS One. 2018; 13(6):e0199111. https://doi.org/ 10.1371/journal.pone.0199111 PMID: 29902248; PubMed Central PMCID: PMC6002122.
- 14. Scott P, Novais FO. Cutaneous leishmaniasis: immune responses in protection and pathogenesis. Nat Rev Immunol. 2016; 16(9):581–92. https://doi.org/10.1038/nri.2016.72 PMID: 27424773.
- 15. Belkaid Y, Mendez S, Lira R, Kadambi N, Milon G, Sacks D. A natural model of Leishmania major infection reveals a prolonged "silent" phase of parasite amplification in the skin before the onset of lesion

formation and immunity. J Immunol. 2000; 165(2):969–77. https://doi.org/10.4049/jimmunol.165.2.969 PMID: 10878373.

- Biyani N, Singh AK, Mandal S, Chawla B, Madhubala R. Differential expression of proteins in antimonysusceptible and -resistant isolates of Leishmania donovani. Mol Biochem Parasitol. 2011; 179(2):91–9. https://doi.org/10.1016/j.molbiopara.2011.06.004 PMID: 21736901.
- Saxena A, Worthey EA, Yan S, Leland A, Stuart KD, Myler PJ. Evaluation of differential gene expression in Leishmania major Friedlin procyclics and metacyclics using DNA microarray analysis. Mol Biochem Parasitol. 2003; 129(1):103–14. https://doi.org/10.1016/s0166-6851(03)00100-2 PMID: 12798511.
- Alves-Ferreira EV, Toledo JS, De Oliveira AH, Ferreira TR, Ruy PC, Pinzan CF, et al. Differential Gene Expression and Infection Profiles of Cutaneous and Mucosal Leishmania braziliensis Isolates from the Same Patient. PLoS Negl Trop Dis. 2015; 9(9):e0004018. https://doi.org/10.1371/journal.pntd.0004018 PMID: 26366580; PubMed Central PMCID: PMC4569073.
- McGwire BS, Satoskar AR. Leishmaniasis: clinical syndromes and treatment. QJM. 2014; 107(1):7–14. https://doi.org/10.1093/qjmed/hct116 PMID: 23744570; PubMed Central PMCID: PMC3869292.
- Remadi L, Haouas N, Chaara D, Slama D, Chargui N, Dabghi R, et al. Clinical Presentation of Cutaneous Leishmaniasis caused by Leishmania major. Dermatology. 2016; 232(6):752–9. https://doi.org/10. 1159/000456543 PMID: 28253508.
- Al-Jawabreh A, Diezmann S, Muller M, Wirth T, Schnur LF, Strelkova MV, et al. Identification of geographically distributed sub-populations of Leishmania (Leishmania) major by microsatellite analysis. BMC Evol Biol. 2008; 8:183. <u>https://doi.org/10.1186/1471-2148-8-183</u> PMID: <u>18577226</u>; PubMed Central PMCID: PMC2447845.
- Banuls AL, Bastien P, Pomares C, Arevalo J, Fisa R, Hide M. Clinical pleiomorphism in human leishmaniases, with special mention of asymptomatic infection. Clin Microbiol Infect. 2011; 17(10):1451–61. https://doi.org/10.1111/j.1469-0691.2011.03640.x PMID: 21933304.
- 23. Pace D. Leishmaniasis. J Infect. 2014; 69 Suppl 1:S10–8. https://doi.org/10.1016/j.jinf.2014.07.016 PMID: 25238669.
- Kopf M, Brombacher F, Kohler G, Kienzle G, Widmann KH, Lefrang K, et al. IL-4-deficient Balb/c mice resist infection with Leishmania major. J Exp Med. 1996; 184(3):1127–36. <u>https://doi.org/10.1084/jem.</u> 184.3.1127 PMID: 9064329; PubMed Central PMCID: PMC2192785.
- Noben-Trauth N, Kropf P, Muller I. Susceptibility to Leishmania major infection in interleukin-4-deficient mice. Science. 1996; 271(5251):987–90. https://doi.org/10.1126/science.271.5251.987 PMID: 8584936.
- Hondowicz B, Scott P. Influence of host and parasite factors on the innate immune response and Th2 stability following infection with Leishmania major. Microbes Infect. 1999; 1(1):65–71. <u>https://doi.org/10.1016/s1286-4579(99)80016-1</u> PMID: 10847768.
- Gupta G, Oghumu S, Satoskar AR. Mechanisms of immune evasion in leishmaniasis. Adv Appl Microbiol. 2013; 82:155–84. https://doi.org/10.1016/B978-0-12-407679-2.00005-3 PMID: 23415155; PubMed Central PMCID: PMC3697132.